Author's response to reviews

Title: Does clinical examination aid in the diagnosis of urinary tract infections in women? A systematic review

Authors:

David Medina-Bombardó (dbombardo@gmail.com)
Antoni Jover-Palmer (ajover@ibsalut.caib.es)

Version: 6 Date: 7 September 2011

Author's response to reviews: see over
Dear Editor:

Thank you for letting us publish in BCM Family Practice.

We have proceeded to modified in the last proof your format recommendations for the manuscript.

Major revisions (we require the author to make these changes)

- Title: please ensure that the title is identical in the manuscript and the submission system. **Done!**

- Abstract: Please ensure that the abstract is identical in the manuscript file on the online submission system. **Done!**

- References: Please make sure references are cited in numerical order. We notice that many of the references have not been cited within the main text, all references must be cited or removed. Please make sure references continue to be cited in numerical order once these references have been cited. **Done!** All references are cited in numerical order in the text or in figures and tables. By mistake, the references in Figure 1 were not included. Now there have been included.

- References: please provide a functioning link to website http://www.hrc.es/investigacion/metadisc.html). **Done!**

- Tables: All tables must consist of both columns and rows, if this is not possible the information can be placed in the manuscript as indented text. **Done!**

- Tables: please note that we are unable to display text vertically within the cells: please re-layout your table without these vertical elements. **Done!**

- Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. **Done!**

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

- Author list: Please place links to the affiliations in numerical order. ?

- Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. **Done!**. We have put all text in black (some paragraphs were blue).
• Tables: please remove the visible vertical lines from your tables. **Done!**

• Tables: Please remove the tables' legends from within the table cells. Place the legends below their respective tables. **Done!**

• Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements. **Done!**

• Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. **Reviewed! We have not seen any typographical mistake.**

Sincerely

David Medina-Bombardó
Antoni J. Jover-Palmer